SHARING LIVES, DIVIDING ASSETS

With many couples separating each year, the question of how to determine the financial and property consequences of such separation has always been a problem area within family law. Should the principles be the same for married and cohabiting couples? Should the division of assets reflect the parties' own expectations or norms imposed by society? These are just two of the questions which the essays in this collection seek to explore.

Recent cases in the House of Lords have seen willingness on the part of the judges to seek out empirical studies to inform their deliberations, but if the law is to engage with empirical data then much more information is needed, both about the arrangements people make during their relationships, and about the impact of the law when a relationship breaks down. This inter-disciplinary work brings together leading academics in the fields of law, economics, sociology and psychology in an attempt to provide some of the missing empirical information. Part I sets out the legal framework and identifies the importance of empirical studies for this area. Part II examines how couples—whether cohabitants or spouses—manage their money during their relationships. Part III then considers the impact that the law currently has on separating couples—examining how legal principles translate into reality and what their consequences are for the parties. Finally, Part IV considers the issue of legal rationality: it may be rational for the law to be shaped by patterns of behaviour, but how far will individual couples allow their behaviour to be shaped by the law?

Sharing Lives, Dividing Assets

An Inter-Disciplinary Study

Edited by
Joanna Miles
and
Rebecca Probert



Published in North America (US and Canada) by
Hart Publishing
c/o International Specialized Book Services
920 NE 58th Avenue, Suite 300
Portland, OR 97213-3786
USA
Tel: +1 503 287 3093 or toll-free: (1) 800 944 6190
Fax: +1 503 280 8832

Fax: +1 503 280 8832 E-mail: orders@isbs.com Website: http://www.isbs.com

© The editors and contributors severally, 2009

The editors and contributors have asserted their right under the Copyright, Designs and Patents Act 1988, to be identified as the authors of this work.

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without the prior permission of Hart Publishing, or as expressly permitted by law or under the terms agreed with the appropriate reprographic rights organisation. Enquiries concerning reproduction which may not be covered by the above should be addressed to Hart Publishing at the address below.

Hart Publishing Ltd, 16C Worcester Place, Oxford, OX1 2JW Telephone: +44 (0)1865 517530 Fax: +44 (0)1865 510710 E-mail: mail@hartpub.co.uk Website: http://www.hartpub.co.uk

> British Library Cataloguing in Publication Data Data Available

> > ISBN: 978-1-84113-259-4

Typeset by Compuscript Ltd, Shannon Printed and bound in Great Britain by CPI Antony Rowe Ltd, Chippenham, Wiltshire

Preface

The idea for this project, and so this book, originated in our growing awareness of two things. First, that the courts' decisions about appropriate basis for dividing wealth on divorce or allocating ownership of property on separation were being increasingly influenced by how parties (are perceived to) organise their lives while together and by what that was supposed to indicate, for example, about their intentions regarding ownership or about the economic impact of relationship breakdown, but that the judges were not always apprised of all the research data necessary to make a fully informed decision about those issues. Secondly, that there is a vast social science literature touching on many issues relevant to legal decision-making in this area, but that researchers in those other disciplines generally run along parallel tracks that never meet each other—or lawyers. We thought that we could usefully try to make a contribution on both fronts by bringing together people working in a wide range of disciplines pertinent to the broad theme of money, property, relationships and relationship breakdown, in order to stimulate debate and cross-fertilisation of ideas between them, and to bring their valuable research to the attention of a wider audience.

The project, run under the aegis of the Cambridge Socio-Legal Group, based at the University of Cambridge, adopted a workshop model. Invited contributors circulated draft chapters on their research in advance of a two-day meeting at which each contributor was then able, following a short presentation, to discuss their work with the whole group. This methodology proved highly successful in enhancing inter-disciplinary understanding, not least in identifying and unravelling areas of misunderstanding, often generated simply by the use of seemingly familiar language in an unfamiliar way. It was also highly productive of ideas for new lines of inquiry in all of the disciplines represented at the workshop, thanks to the fresh perspectives cast on otherwise familiar territory by those whose disciplinary background provided a different vantage point from which to view the material. In light of those discussions, the contributors were then able to revise their chapters before publication, and our introductory chapter benefited considerably from insights offered by members of the group during our discussions over those two days. We hope that the result is a volume that provides a valuable resource, giving access to findings on wide range of issues from a wide range of disciplines, in a way that will both inform legal development in this field and encourage further research.

We are extremely grateful to the Cambridge Socio-Legal Group, the John Hall Fund of the University of Cambridge, and to Trinity College, Cambridge for their financial support for this project, and to the staff of Trinity College who helped the workshop to run so smoothly. In addition to the contributors themselves,

vi Preface

whose biographies appear in the following pages, we must also acknowledge a huge debt to the following individuals for their contributions to discussion during the workshop: Belinda Brooks-Gordon, Ceridwen Roberts, Martin Richards, Brian Sloan and Anke Zimmermann. Finally, we must also thank Liam D'Arcy Brown for his assistance with proof-reading and editing.

Contents

	efacev otes on Contributorsix
	Part I: General Issues
1.	Sharing Lives, Dividing Assets: Legal Principles and Real Life
2.	The Perils of Reforming Family Law and the Increasing Need for Empirical Research, 1980–2008
	Part II: Work, Money and Property within Intimate Relationships: Expectations and Actions
3.	Paid and Unpaid Work: Can Policy Improve Gender Inequalities?41 JACQUELINE SCOTT AND SHIRLEY DEX
4.	Managing Money in Intimate Relationships: Similarities and Differences between Cohabiting and Married Couples
5.	Financial Practices in Cohabiting Heterosexual Couples: A Perspective from Economic Psychology
6.	The Role of Personal Relationships in Borrowing, Saving and Over-indebtedness: A Life-course Perspective
	Part III: Dividing the Assets on Relationship Breakdown
7.	Money, Property, Cohabitation and Separation: Patterns and Intentions139 GILLIAN DOUGLAS, JULIA PEARCE AND HILARY WOODWARD
8.	Financial Arrangements on the Breakdown of Cohabitation: Influences and Disadvantage
9.	Chaos or Consistency? Ancillary Relief in the 'Everyday' Case

viii Contents

10.	Self-determination or Judicial Imposition? Translating the Theory into Practice
11.	Who Wins, Who Loses and Who Recovers from Divorce?
12.	Pension Accumulation and Gendered Household Structures: What are the Implications of Changes in Family Formation for Future Financial Inequality?
	Part IV: A Rational Approach?
13.	Rational Decision-making and Intimate Cohabitation
14.	Legal Rationality and Family Property: What has Love got to do with it?303 ANNE BARLOW
Ind	ex

Notes on Contributors

Anne Barlow studied Law with French at the University of Sussex and is Professor of Family Law and Policy at the University of Exeter. The main focus of her research has been on family and housing law, and especially cohabitation law, and she has published widely in these fields. In particular, she is author of *Cohabitants and the Law* (Butterworths, 2001) and (with S Duncan, G James, and A Park) of *Cohabitation, Marriage and the Law* (Hart Publishing, 2005). She has, since 2000, directed and co-directed a number of empirical socio-legal projects funded by the Nuffield Foundation, the Ministry of Justice and the Economic and Social Research Council (ESRC) including 'Family Restructuring, the Common Law Marriage Myth and the Need for Legal Realism, Community of Property—A Regime for England and Wales?' and 'The Common Law Marriage Myth and Cohabitation Law Reform Revisited'. She has been consulted by the Law Commission for England and Wales, the Scottish Executive, and the Office of Law Reform, Northern Ireland on reform of cohabitation law and the implications of her empirical research for this process.

Carole Burgoyne is Senior Lecturer in the School of Psychology, University of Exeter. Her main research interests are in economic psychology and include studies of money, marriage, currency change and distributive justice. Her most recent projects are 'The Common Law Marriage Myth Revisited' (led by Anne Barlow); 'Financial Management Practices in Non-traditional Heterosexual Couples' (with Stefanie Sonnenberg and Anne Barlow); 'Beliefs about the Allocation of Money in Marriage' (with Stefanie Sonnenberg); and 'Money management in Lesbian and Gay Couples' (with Victoria Clarke).

Shirley Dex is Professor of Longitudinal Social Research in the Centre for Longitudinal Studies, Institute of Education, University of London. She has held posts at the Judge Business School, University of Cambridge, the Institute for Social and Economic Research at the University of Essex, and the Economics Department at the University of Keele. From 1998 to 2003 she was Research Advisor to the Joseph Rowntree Foundation's Work and Family Life Programme. She has published widely on women's employment and cross-national comparative research, equal opportunities, families and work, ethnic minorities and employment, flexible working arrangements in organisations, work and care, and family policy. Recent books include *Families and Work in the 21st Century* (Joseph Rowntree Foundation, 2003). She has also co-edited (with Heather Joshi) *Children of the 21st Century* (The Policy Press, 2005) and (with Jacqueline Scott and Heather Joshi) *Women and Employment: Changing Lives and New Challenges* (Edward Elgar Publishing Ltd, 2008).

Antony Dnes currently holds a chair in Economics at the University of Hull, and has held previous appointments at the Universities of Birmingham, Edinburgh and St Andrews in the UK, and at the Virginia Tech and the George Mason Law School in the US. He completed his PhD in Economics at the University of Edinburgh, and has an LLB from the University of London. He is a specialist in the economic analysis of law, with a number of published articles dealing with family topics, in both law and economics journals. He co-edited *The Law and Economics of Marriage and Divorce* (CUP, 2002), a collection of articles from leading academics on both sides of the Atlantic.

Gillian Douglas obtained her LLB at the University of Manchester in 1977 and her LLM at the London School of Economics in 1978. She is currently Professor of Law and Head of Cardiff Law School, having previously taught at the University of Bristol and the National University of Singapore. She is the co-editor of the Child and Family Law Quarterly and of the Case Reports section of Family Law. Her publications include (with Nigel Lowe) Bromley's Family Law (9th and 10th editions) and An Introduction to Family Law (Clarendon Law Series, 1st and 2nd editions). With colleagues from disciplines including psychology and social work she has conducted a number of empirical studies in law, including How Parents Cope Financially on Marriage Breakdown (2000, funded by the Joseph Rowntree Foundation), Grandparenting in Divorced Families (2004, funded by the Nuffield Foundation), and A Failure of Trust: Resolving Property Disputes on Cohabitation Breakdown (2007, funded by the ESRC).

Sally Dowding is a District Judge based at Birmingham Civil Justice Centre. She obtained an LLB degree from the University of Manchester in 1973 and subsequently qualified as a solicitor. For most of a practising career spanning nearly 30 years she specialised in family law. She was a member of the Law Society's Family Law Committee and Children Law Sub-committee for the 10 years preceding her judicial appointment and chaired the latter committee for nearly four years.

John Eekelaar is Emeritus Fellow of Pembroke College, Oxford, and a Fellow of the British Academy. He has written and conducted research in family law since the 1970s, and is currently Joint Director of the Oxford Centre for Family Law and Policy with Mavis Maclean, with whom he has co-authored several books. His latest book, *Family Law and Personal Life*, was published by Oxford University Press in hardback in 2006, and paperback in 2007. He also co-edits the *International Journal of Law, Policy and the Family*.

Andrea Finney is a Research Fellow in the Personal Finance Research Centre at the University of Bristol. Primarily a quantitative researcher, she previously spent six years at the Home Office and, latterly, the Office for National Statistics, where she was involved in rolling-out the new longitudinal survey of household assets. Andrea's current research interests lie in furthering understanding of saving and borrowing behaviour, financial vulnerability and decision-making. She is co-author of reports on over-indebtedness (for the European Commission),

the results from the baseline survey of consumer purchasing (for the Financial Services Authority) and modelling exposure to financial risk (for Genworth Financial).

Hayley Fisher is a PhD student in the Faculty of Economics at the University of Cambridge, supervised by Thomas F Crossley and Hamish Low. Her research interests include family and labour economics, and she is currently considering the economic impact of divorce legislation on the decision to marry or to cohabit.

Emma Hitchings is Senior Lecturer in the School of Law, University of Bristol. She studied law as an undergraduate before taking her PhD in 2003. Her main research interests lie in the field of family law, where she has written and worked on issues concerning domestic violence, adoption, ancillary relief, and the recognition of adult relationships, particularly same-sex couples. She is case commentaries editor for the Journal of Social Welfare and Family Law and an academic member on the Family Law Committee of the Law Society of England and Wales. She is currently engaged in research on pre-nuptial agreements for the Law Commission for England and Wales.

Jane Lewis is Director of Research, Evidence and Evaluation at the National Children's Bureau and was formerly Director of the Qualitative Research Unit at the National Centre for Social Research. She practised as a solicitor before moving into social policy research. Her recent research includes Separating from Cohabitation: Making Arrangements for Finances and Parenting (2006, funded by the Department for Constitutional Affairs); Settling Up: Making Financial Arrangements after Divorce or Separation (2002, funded by the Nuffield Foundation), and Pensions and Divorce: A Qualitative Study among Solicitors (2000, funded by the Department for Work and Pensions). She has written widely on research methods and is co-editor of Qualitative Research Practice, published by Sage in 2003. She has a long-standing interest in relationship breakdown and more broadly in research on children and young people's lives; social inclusion, inequality and disadvantage; and research methods.

Hamish Low is a Senior Lecturer in the Faculty of Economics at the University of Cambridge and a Fellow of Trinity College, Cambridge. He is also a research fellow at the Institute for Fiscal Studies. His main research interests are in understanding the risks that individuals face over their lifetimes and in analysing the extent to which these risks are insured. A particular interest has been in understanding changing patterns of women's labour supply across different generations. He has published primarily in economic journals, such as the American Economic Review and the Journal of Public Economics.

Mavis Maclean CBE is Joint Director with John Eekelaar of the Oxford Centre for Family Law and Policy, Department of Social Policy and Social Work, University of Oxford. She also acts as Academic Adviser to the Ministry of Justice. Recent publications include Parenting after Partnering (Hart Publishing, 2007), Cross Currents in Family Law, edited with Katz and Eekelaar (OUP, 2000), and Family Lawyers with Eekelaar and Beinart (Hart, 2000). She is a former President of the Research Committee for the Sociology of Law and a Fellow of the International Institute for Sociology of Law, Onati, Spain.

Joanna Miles is a University Lecturer in the Faculty of Law and Fellow of Trinity College, University of Cambridge. Her research interests lie mainly in the area of family property and financial remedies following relationship breakdown, whether marital or non-marital. Her work has been considered and cited judicially in England and Wales, Ireland, and New Zealand. She was recently seconded to the Law Commission for England and Wales for two years to work as a lawyer on its project 'Cohabitation: The Financial Consequences of Relationship Breakdown'. She is currently embarking on a Nuffield Foundation-funded project examining the first three years' operation of the new remedies between cohabitants in Scotland, introduced by the Family Law (Scotland) Act 2006, with Professor Fran Wasoff at the University of Edinburgh. She is co-author (with Sonia Harris-Short) of Family Law: Text, Cases, and Materials (OUP, 2007).

Julia Pearce graduated from the University of Bristol in 1988 with a degree in Law, subsequently practising as a solicitor specialising in family law for seven years. She obtained her LLM in 1996, following which she left private practice to work full-time in socio-legal research. Joining the Law Department of Bristol University she has worked on a number of research studies in family law with colleagues in the Schools of Law and of Social Policy. These include A Study of Children Act Applications (1998, funded by the Nuffield Foundation), An Evaluation of the Family Mediation Pilot Scheme (1999, for the Legal Services Commission), A Survey of Ancillary Relief Outcomes (1999, for the Lord Chancellor's Department), and A Failure of Trust: Resolving Property Disputes on Cohabitation Breakdown (2007, funded by the ESRC).

Debora Price is a Lecturer in Social Policy at the Institute of Gerontology at King's College London, where she specialises in the study of pensions and the poverty of older people, issues relating to household money over the life-course, and the legal regulation of the financial consequences of family formation and dissolution. She also has an interest in survey methodology and data analysis, regularly undertaking consultancy work in this area. Formerly a barrister specialising in family law, her ESRC PhD research focused on the impact of family change on pension scheme participation in the UK. She has recently concluded an ESRC/DWP fellowship placement at the Department for Work and Pensions reviewing measures of pensioner poverty. Her current research includes the ESRC-funded project 'Behind Closed Doors: Older Couples and the Management of Household Money'.

Rebecca Probert is a Senior Lecturer at the University of Warwick, teaching family law, child law, and a course on 'law and the intact family'. She has written widely on the topics of cohabitation and marriage, particularly from a historical perspective. Her recent publications include Marriage Law and Practice in the Long Eighteenth Century: A Reassessment (CUP, 2009) and 'Looking back on the overlooked: cohabitants and the law 1857–2007' in N Lowe and G Douglas (eds) The Continuing Evolution of Family Law (Jordans, 2009). She is also co-author (with J Masson and R Bailey-Harris) of Cretney: Principles of Family Law (Sweet & Maxwell, 2008), editor of Family Life and the Law: Under One Roof (Ashgate, 2007), and co-editor (with Stephen Gilmore and Jonathan Herring) of Responsible Parents and Parental Responsibility (Hart Publishing, 2009).

Jacqueline Scott is Professor in Empirical Sociology at the University of Cambridge and a Fellow of Queens' College. She directs the ESRC's Research Network on Gender Inequalities in Production and Reproduction. Her former positions include Director of Research at the ESRC Centre for Micro Social Change at the University of Essex, where she was involved in the original design and implementation of the British Household Panel Survey. Her research interests include life-course research; gender-role change; attitudinal research; and ageing and well-being. She has recently co-edited (with Judith Treas and Martin Richards) the Blackwell Companion to the Sociology of Families (Blackwell, 2004); (with Yu Xie) the Sage Benchmark Series on Quantitative Sociology (Sage, 2005); and (with Shirley Dex and Heather Joshi) Women and Employment: Changing Lives and New Challenges (Edward Elgar Publishing Ltd, 2008).

Stefanie Sonnenberg is a Lecturer in Psychology at the University of Portsmouth. She graduated from the University of Exeter with an MSc in Economic Psychology (1998) and completed her PhD in Social Psychology at the University of St. Andrews (2004). Her general research interest lies at the intersection between economic psychology and social psychology, and her work primarily focuses on the relationship between identity and economic practices. Her recent projects (with Carole Burgoyne and Anne Barlow) have examined financial management practices in intimate relationships.

Jean Taylor is an Associate Fellow at the Office for Public Management (OPM). Prior to joining OPM, Jean was a Senior Researcher at National Centre for Social Research (NatCen), where she undertook fieldwork for, and co-authored, Separating from Cohabitation: Making Arrangements for Finances and Parenting (2006, funded by the Department for Constitutional Affairs). Jean holds an MSc in European Social Policy Analysis from the University of Bath, and a BA in Modern History from the University of Oxford.

Rosalind Tennant is a Research Director at NatCen. She specialises in qualitative research and has worked across a range of social policy areas including families and work, older people's services, and social inclusion. She is co-author of the research report Separating from Cohabitation: Making Arrangements for Finances and Parenting (2006, funded by the Department for Constitutional Affairs). Her current work focuses on the areas of learning, skills, and employment,

xiv Notes on Contributors

in particular special educational needs. Rosalind graduated in 2000 with a BA in Communication and in 2001 with a MA in Research in Language and Communication, both from the University of Wales, Cardiff. She worked in qualitative and quantitative market research for a number of years before joining NatCen's Qualitative Research Department in 2004.

Carolyn Vogler is Reader in Sociology at City University, London. She has published extensively in the areas of globalisation, social class, unemployment, gender segregation in the labour market, and money and gender relations within intimate relationships. Currently, her main research interests are in the area of money, power, and inequality in new forms of intimate relationships.

Hilary Woodward graduated from the London School of Economics with a degree in Sociology and Social Anthropology in 1967. She worked as a research associate on projects with the Institute of Education, the Institute of Psychiatry and the Medical Research Council, before taking up training in law. She practised as a family lawyer from 1982 to 2003 and was accredited as a Resolution cohabitation specialist in 2001. She has worked as a family mediator with Bristol Family Mediation since 1993, and joined Cardiff Law School as a research associate in 2003. She is co-author of 'Ancillary Relief Outcomes' (2000) 12 *Child and Family Law Quarterly* 43, *Divorce for Dummies* (John Wiley and Sons Ltd, 1st and 2nd editions) and *A Failure of Trust: Resolving Property Disputes on Cohabitation Breakdown* (2007, funded by the ESRC).

Part I General Issues

Sharing Lives, Dividing Assets

Legal Principles and Real Life

JOANNA MILES AND REBECCA PROBERT

INTRODUCTION

PERENNIAL DEBATE in family law concerns the principles that should be applied in determining the financial and property consequences of relationship breakdown. Key questions include the following: should the principles (whatever they may be) be the same for married¹ and cohabiting couples? What principles should be adopted, for both or for either category of relationship? Should the division of resources, and so the applicable principles, reflect the parties' own expectations or societal norms? If the former, how are those expectations or intentions to be discerned? If the latter, how should policy-makers—whether judicial or legislative—determine what those norms are or should be?

The House of Lords has endeavoured to take heed of empirical data in its deliberations on some of these questions in recent cases concerning the financial consequences of divorce² and the separation of cohabitants.³ The Law Commission similarly drew on socio-legal research⁴ in devising its recommendations for reform of the law governing cohabitants' financial remedies following separation.⁵ More recently, it cited the *lack* of large-scale empirical research that would provide a contemporary socio-economic background for law reform amongst its reasons for declining to take on a project considering the financial remedies available on the dissolution of marriage and civil partnership.⁶ The importance of both quantitative and qualitative socio-legal research in informing the development and reform of family law is therefore widely accepted. But if family lawyers and policy-makers

¹ The same questions apply to civil partners, although, for the reasons set out below, the focus of this collection is on opposite-sex couples.

² Miller v Miller, McFarlane v McFarlane [2006] UKHL 24 (referred to as 'Miller; McFarlane').

³ Stack v Dowden [2007] UKHL 17.

⁴ For example research concerning couples' money-management practices and public attitudes towards law and law reform in this area: Vogler (2005); Burgoyne et al (2006); Vogler, Brockmann and Wiggins (2006, 2008).

⁵ Law Commission (2006, 2007).

⁶ Law Commission (2008) [5.6].

are to engage successfully with empirical data in this sphere then, as the Law Commission's recent decision about its future programme of work indicates, much more research is needed, not least about the arrangements people make during their relationships and about the operation and impact of the current law.

This inter-disciplinary volume is intended to make some contribution towards addressing this gap in our knowledge—and to overcome perceived difficulties of accessing material from what might otherwise (to the legal audience, in particular) be an unfamiliar literature—by bringing together leading academics in the fields of law, economics, sociology, psychology and other social sciences. It seeks to provide a critical examination of issues relating to the financial relationships and obligations between couples (both during their relationships and on separation) by collating empirical research conducted from the perspectives of various social science disciplines.

Part I of the book, of which this chapter forms a part, sets out the legal framework and identifies the importance of empirical studies for this area. If legal principles are to take account of what couples do in practice, then information about practice is essential. It is therefore crucial to build up a body of research on the ways in which couples behave and, very importantly, what meaning can reliably be attributed to that behaviour in order to inform what will otherwise be 'instinctive', but quite possibly incorrect, assessments of these matters. Part II accordingly examines how couples, whether cohabitants or spouses, manage their various resources—monetary, property and human—during their relationships. Part III considers the impact that the law currently has on separating couples, examining how legal principles translate into reality and what their consequences are for the parties. This requires separate consideration of cohabitants and spouses, since significantly different rules currently apply to each group. Finally, Part IV considers the issue of legal rationality: it may be rational for the law to be shaped by patterns of behaviour, but how far will individual couples allow their behaviour to be shaped by the law?

The focus on opposite-sex couples is both necessary and justified: necessary, because much⁷ of the research carried out to date has focused exclusively on married couples and opposite-sex cohabitants, and one cannot assume that it can be transposed to couples of the same sex (and still less to those who are not in couple relationships⁸); and justified, because one of the most striking findings

⁷ There is a growing literature on the family practices and perspectives of same-sex couples (see eg Weeks, Heaphy and Donovan (2001); Smart, Mason and Shipman (2006), Clarke, Burgoyne and Burns (2007); Burns, Burgoyne and Clarke (2008); Peel and Harding (2008)). The data is still limited in comparison to that available for opposite-sex couples, inevitably so for civil partnership, given its recent introduction.

⁸ Although the position of wider kin has attracted more attention in recent months in the wake of *Burden and Burden v UK* (2008) 47 EHRR 38, and the proposals contained in the consultation paper issued by Resolution and Lord Lester in 2008 (see www.resolution.org.uk/livingtogether/). For literature in this area, see eg the work of Roseneil and others on friendship and non-conventional partnerships within the CAVA research project: www.leeds.ac.uk/cava/research/strand3d.htm; Williams (2004); Bottomley and Wong (2006).

that permeates all of the chapters is the continuing relevance of gender to the division of both responsibilities and wealth, as Scott and Dex discuss in chapter three. This has implications for the way in which money is dealt with during the relationship and the resources available to each party, as illustrated by Vogler (chapter four), and Burgoyne and Sonnenberg (chapter five) in their examinations of money management practices, and by Finney (chapter six), focusing on the issue of debt. Crucially, it also has a profound impact upon relationship breakdown, both in the immediate aftermath, as Lewis, Tennant and Taylor show in their discussion of disadvantage on the breakdown of cohabiting relationships in chapter eight; and in the longer term—quite how long being revealed by Fisher and Low, in their analysis of post-separation income levels in chapter eleven, and by Price, in her discussion of pension assets in later life in chapter twelve.

THE LEGAL AND POLICY CONTEXT

The laws currently applicable to financial disputes that may arise between spouses and cohabitants are complex and, in key respects, different. During the currency of these relationships, much of the law applies even-handedly to both types of relationship. While spouses enjoy certain limited statutory entitlements (notably statutory rights to occupy the shared home⁹), they are for the most part treated in private law¹⁰ as entirely separate individuals, each equally entitled to earn, acquire and own property, and to manage those resources independently of the other party. 11 Indeed, they are formally treated no differently from cohabitants. The same rules of property and trust law apply across the board in order to determine who owns what—in particular, who owns the family home and in what shares (if any). The legal position of spouses and cohabitants diverges substantially, however, at the point of relationship breakdown. While both may apply for the courtbased remedy of tenancy transfer, 12 only spouses have access to a more general jurisdiction which enables the discretionary redistribution of income and capital resources, present and future.¹³ Cohabitants, by contrast, are left with whatever the law of property and trust entitles them to, supplemented (or reduced) only by

⁹ Family Law Act 1996 ss 30-34. See also the statutory duty to maintain during marriage (in practice substantially less important than maintenance following divorce): eg Matrimonial Causes Act 1973 s 27. Other spousal rights and duties are of rather less practical relevance in the current social context, and have not been extended to civil partners: Married Women's Property Act 1964 s 1 (determining the ownership of savings made by the wife (specifically) from a housekeeping allowance); the 'presumption of advancement' whereby transfers of wealth made from husband to wife (but not vice versa) are presumed (now only weakly—Pettitt v Pettitt [1970] AC 777) to be intended as a gift. See generally Harris-Short and Miles (2007) ch 3; Masson, Bailey-Harris and Probert (2008) chs 3 and 4.

¹⁰ Contrast the aggregation of spouses' (and cohabitants') resources for the purpose of calculating entitlement to means-tested benefits and tax credits.

¹¹ Since the Married Women's Property Acts of the late 19th century.

¹² Family Law Act 1996 s 53 and Sch 7.

¹³ Matrimonial Causes Act 1973 Part II.

whatever is afforded (or demanded) by the law relating to the financial support of children. ¹⁴ We outline the details of these laws further below.

It is likely that many people are surprised by these key features of the current law. For example, we know from the British Social Attitudes survey that a majority (albeit shrinking) still positively believe (in general terms¹⁵) that spouses and cohabitants enjoy similar legal status, leaving many cohabitants disappointed (and seriously disadvantaged) on relationship breakdown. Conversely, a large number of spouses would probably be surprised to discover that they do not automatically—by virtue of marriage—enjoy joint ownership of all or certain assets (particularly the matrimonial home and/or its essential household contents), as is the case in certain European jurisdictions operating systems of 'community of property'.

The Law of 'Ancillary Relief' on Divorce

Upon divorce, the court has a discretion as to how it should exercise its wideranging powers to reallocate assets between the former spouses. Until 1984, courts were directed to exercise these powers in such a way as would keep the parties, so far as it was practicable and (having regard to their conduct) just to do so, in the financial position in which they would have been if the marriage had not broken down. Since the removal of that direction by the Matrimonial and Family Proceedings Act 1984 (see further Maclean and Eekelaar, chapter two, this volume), there has been no statutory guidance as to the aims of the process. The legislation simply states that first consideration is to be given to the welfare of any child of the family, that the court is to take all the circumstances of the case into account (referring judges to a list of specific but non-exhaustive factors), and that the court is to consider the desirability of an order that will achieve a clean break between the parties, that is to say, to bring to an end any ongoing financial ties between them. In line with the shift away from a divorce law based on fault to one based on the breakdown of the marriage, the conduct of the parties is rarely a factor that will influence the division of the parties' assets, save where it is 'obvious and gross' 18 or directly affects the assets available for division. 19

¹⁴ Child Support Act 1991, and Children Act 1989 s 15 and Sch 1.

¹⁵ Contrast the responses given to more specific questions regarding particular legal entitlements: Barlow et al (2001).

¹⁶ See Barlow et al (2008).

¹⁷ See generally the attitudes of those interviewed by Cooke, Barlow and Callus (2006) about what the law *should* be, both for spouses and cohabitants; a majority of British Social Attitudes respondents think that *cohabitants* should pool their resources: Barlow et al (2008) table 2.4; see Harris-Short and Miles (2007) 3.5 for discussion of community of property and joint ownership systems.

¹⁸ See Wachtel v Wachtel (No 2) [1973] Fam 72.

¹⁹ As in the case of financial or litigation misconduct: see eg M v M (Financial Misconduct: Subpoena Against Third Party) [2006] 2 FLR 1253.

The principles underpinning the court's exercise of discretion were discussed by the House of Lords in the cases of White v White²⁰ and Miller; McFarlane²¹ and subsequently by the Court of Appeal in Charman v Charman. 22 Three bases for reallocation were articulated: need, compensation, and sharing. From a theoretical perspective this proliferation poses a problem, in that each of the bases presupposes a slightly different model of marriage; from both a practical and a theoretical perspective, there is the challenge of deciding whether, and if so when, needs should take precedence over sharing, and how compensation fits into the court's assessment. There have also been debates as to what exactly is to be shared: does the principle apply only to assets acquired as a result of the parties' joint efforts during the marriage, or does it extend to other assets acquired before,²³ during, or after²⁴ this period?²⁵

At present the position would appear to be that the court will first consider the needs of the parties, and that any assets owned by either party, whenever acquired, may be used to ensure that such needs are met.²⁶ In lower-income cases, meeting the needs of the parent with primary care for the children of the marriage may well result in that party receiving more than half of the assets. If there is a surplus after the needs of the parties are met, then the sharing principle comes into play. This principle does not, however, simply dictate that all assets, or even all assets acquired during the marriage, are to be divided equally between the parties: the duration of the marriage and the contributions of the parties may suggest that a different distribution would be fairer.²⁷ As the Court of Appeal put it in *Charman* v Charman, 'property should be shared in equal proportions unless there is good reason to depart from such proportions'.²⁸

This flexible approach reflects the way in which the modern courts have consistently eschewed a formulaic approach.²⁹ They have also been reluctant to develop the idea of compensation introduced in Miller; McFarlane. While there is now a greater awareness of the impact of sacrificing a career, 30 such sacrifices tend to be addressed by means other than including a specifically compensatory element

²⁰ [2001] 1 AC 596.

²¹ [2006] UKHL 24.

²² [2007] EWCA Civ 503.

²³ A clear answer in the negative was given in B v B (Ancillary Relief) [2008] EWCA Civ 543; see also McCartney v Mills McCartney [2008] EWHC 401 (Fam).

²⁴ See eg the debates regarding post-separation property in *Rossi v Rossi* [2006] EWHC 1482 (Fam); S v S [2006] EWHC 2339 (Fam); H v H [2007] EWHC 459 (Fam).

²⁵ See also Vaughan v Vaughan [2007] EWCA Civ 1085, in which Wilson LJ drew a distinction between equality of assets and equality of outcome.

²⁶ See eg S v S (Non-Matrimonial Property: Conduct) [2006] EWHC 2793; S v S (Ancillary Relief: Importance of FDR) [2007] EWHC 1975.

²⁷ See eg S v S (Non-Matrimonial Property: Conduct) [2006] EWHC 2793; NA v MA [2006] EWHC 2900; McCartney v Mills McCartney [2008] EWHC 401 (Fam).

^{28 [2007]} EWCA Civ 503 [65].

²⁹ See eg *H v H* [2007] EWHC 459 (Fam).

³⁰ See eg Lauder v Lauder [2007] EWHC 1227 (Fam); VB v JP [2008] EWHC 112 (Fam).

in that party's share of the assets.³¹ In lower-income cases the ongoing effect of sacrifices made during the marriage (for example giving up paid employment to care for the children of the marriage) will be dealt with as an aspect of that party's needs;³² in higher-income cases the sharing principle may produce more than the spouse would have earned as a result of paid employment, 33 and, or alternatively, the loss of earning capacity will be recognised through a 'generous assessment' of that party's needs.³⁴ In other cases the principle is not seen as relevant, either because the spouse had not developed a career before the marriage, 35 or because giving up work was a 'lifestyle choice' rather than a necessity. 36 And even if the principle is seen as appropriate, courts may struggle with its application to the case before them: one judge suggested that it was neither possible nor desirable to attempt to ascertain the career that the spouse would have had.³⁷ As yet, therefore, there have been few cases in which the concept of compensation has played an independent role in the court's deliberations.

The Law of Property and Trusts: Dividing the Assets when **Cohabitants Separate**

Decisions made during the lifetime of the relationship have a profound, if not necessarily anticipated, effect on the parties' property rights on separation. Most couples who have purchased a home together in the past decade will be jointly registered as legal owners of that property and will have made an express declaration as to their beneficial interests in the property³⁸ (even if some of them may not have realised the effect of so doing³⁹). Such a declaration is determinative of their rights—in the absence of fraud or common mistake—regardless of their respective contributions.⁴⁰ The recent House of Lords decision in Stack v Dowden⁴¹ made clear that if a couple purchase a home in joint names without making any such declaration, the starting point will be that they too intended that the beneficial title should be held jointly,⁴² and strong evidence will be required

- ³¹ See eg *H v H* [2007] EWHC 459 (Fam).
- ³² See eg Lauder v Lauder [2007] EWHC 1227 (Fam).
- ³³ CR v CR [2007] EWHC 3334 (Fam).
- ³⁴ See eg Lauder v Lauder [2007] EWHC 1227; VB v JP [2008] EWHC 112.
- ³⁵ NA v MA [2006] EWHC 2900.
- ³⁶ S v S (Non-Matrimonial Property: Conduct) [2006] EWHC 2793.
- ³⁷ P v P [2007] EWHC 779 (Fam). See also VB v JP [2008] EWHC 112 (Fam) and CR v CR [2007] EWHC 3334 (Fam).
- 38 Since 1998 the relevant Land Registry forms have invited those purchasing a home in joint names to indicate whether they hold the beneficial title as joint tenants, or as tenants in common in equal shares, or on some other trusts. However, the Land Registry will not reject a form that fails to complete this declaration.
 - ³⁹ See Douglas, Pearce and Woodward, ch 7, this volume.
 - 40 Goodman v Gallant [1986] 1 FLR 513.
 - ⁴¹ [2007] UKHL 17.
 - 42 See also Abbott v Abbott [2007] UKPC 53.

to show that, 'unusually', a different arrangement was intended. By contrast, if the home was purchased in the name of one party (or was owned by him or her before the relationship began), then the onus is on the other party to establish a beneficial interest under the law of trusts. This may be done by showing either that the claimant made a financial contribution to the initial purchase price or that there was an agreement, arrangement or understanding between the parties on which the claimant has relied to his or her detriment.⁴³ The former may give rise to a resulting trust (which will yield an interest in the property proportionate to the original contribution), or, more probably in the domestic context, 44 will lead the court to infer a constructive trust on the basis that there was a common intention that the beneficial interest should be shared, the financial contribution establishing the necessary element of detrimental reliance. The extent of the parties' interests under a constructive trust will be determined by reference to the whole course of dealing between the parties.

Such distinctions lead to some stark differences in outcome. In Fowler v Barron, 45 for example, the property had been conveyed into joint names, but during the 17 years of the parties' relationship it was Mr Barron who made all the payments under the mortgage and who met other household expenses. Miss Fowler's salary was used for more transient items, such as holidays, meals out, and clothes in particular for the children and herself. The court's interpretation of this was that, despite the lack of a joint bank account, the parties had effectively pooled their assets and did not care who paid for what, or how much; the inference was that they had not intended that their different contributions should lead to them having different interests in the property. Each party was thus entitled to 50 per cent of the equity in the property. By contrast, in James v Thomas, 46 Miss James moved into a home already owned by Mr Thomas, and he was reluctant to convey it into their joint names. Although she worked unpaid in his business for the better part of their 15-year relationship and the mortgage was paid from the profits of that business, the court held that there was no common intention that she should have an interest in the property, and that her contributions were simply attributable to the fact that she and Mr Thomas 'were making their life together as man and wife'.47 She was therefore not entitled to any interest in the home.48

Whether these outcomes truly reflected the intentions of the parties is open to question: Mr Thomas acknowledged that it would be fair for Miss James to receive some interest in the property, while Mr Barron argued that, while he had intended that Miss Fowler should inherit the property if he pre-deceased her, he had not intended that she should be entitled to half if they separated. This is not an

⁴³ Lloyds Bank v Rosset [1991] AC 107.

⁴⁴ At least where the parties are sharing a home: see eg Adekunle and Others v Ritchie [2007] EW Misc 5 (EWCC); cf Laskar v Laskar [2008] EWCA Civ 347.

^{45 [2008]} EWCA Civ 377.

^{46 [2007]} EWCA Civ 1212.

⁴⁷ ibid [36] (Chadwick LJ).

⁴⁸ See also *Morris v Morris* [2008] EWCA Civ 257.

unusual view: it has been suggested that many in this situation would be perfectly happy for their partner to inherit all their property in the case of their death, but are far less willing to split their assets equally in case of separation.⁴⁹ The sharing that is part of a loving relationship is often fiercely resisted by the parties when that relationship has come to an end, whether they were married or cohabiting (as is evident from the research of Lewis et al in this volume). The law, however, takes the view that while the parties' intentions regarding the shares each has in the property may change over time, those intentions—and so those shares—must at any given point in time be the same for all purposes.⁵⁰

Financial Provision for Children whose Parents Live Apart

In order to complete the picture of the current law relating to the distribution of wealth on relationship breakdown, it is necessary briefly to outline the law governing provision for children. This law applies almost identically to children of married and unmarried parents.⁵¹ However, given the very different background of provision/division of property between the adults, described above, the overall outcomes for formerly marital and cohabiting households and the children within them are very different.

The Child Support Act 1991, as amended, declares the responsibility of each parent to maintain his children,⁵² regardless of the nature of the parents' relationship. In the case of a 'non-resident parent' (a parent with whom the child does not principally live), this responsibility is met by the payment of child maintenance calculated by reference to a formula which (subject to possible variation on limited grounds set out in the Act) requires payment of a given percentage of that parent's income, depending on the number of children to be supported. Recent changes effected by the Child Maintenance and Other Payments Act 2008 mean that child maintenance is now viewed principally as a matter to be resolved by (non-binding) private agreement between the parties.⁵³ The statutory agency, its formula and enforcement powers are intended to provide a fallback where such agreement is unattainable or breaks down.

Basic child maintenance (generally made by way of regular payments) is not a matter for the courts, except where agreed by the parties and enshrined in a

⁴⁹ See Douglas, Pearce and Woodward, ch 7, this volume.

 $^{^{50}}$ Contrast the findings of Douglas, Pearce and Woodward, ch 7, this volume, who find that parties may have precisely the sort of multiple intention that the law does not accommodate.

⁵¹ The only key distinction pertains to the responsibility of step-parents: only an individual who is married to (or a civil partner of) a child's legal parent may be required to maintain the child; someone who has merely lived with that parent may not: see the concepts of 'parent' in Children Act 1989 Sch 1 para 16(2); and 'child of the family' in Matrimonial Causes Act 1973 s 52(1).

⁵² Child Support Act 1991 s 1. Only 'qualifying children', as defined by ss 3 and 55 of the Act, are owed this duty.

⁵³ Prior to these reforms, if the parent with care was claiming a means-tested benefit, the Child Support Agency's involvement was effectively mandatory.

'consent order'.⁵⁴ However, the courts have exclusive jurisdiction to order various forms of capital provision 'for the benefit of the child'.⁵⁵ Crucially, that qualification has been interpreted in such a way as effectively prevents the outright transfer of property to the parent with care. Generally, any capital transferred must either be spent for the child's benefit during the child's minority (for example on depreciating assets such as a car for taking the child to and from school and other activities, or 'white goods' for the child's home), or, in the case of the transfer of a house, the property must revert to the other parent once the child attains independence.⁵⁶ Provision for dependent children of a formerly married or cohabiting couple can therefore offer no long-term or direct security for the parent with care. But, as we have seen, that limitation of these remedies is of far greater significance for former cohabitants (who have no personal financial remedy and may have no claim in property law) than for former spouses (who have the benefit of financial relief under the matrimonial legislation, whatever their property rights may be).

Future Reform?

As increasing numbers of couples cohabit outside marriage and as more children are born outside marriage, reform providing statutory financial remedies between cohabitants—already available in a number of other jurisdictions⁵⁷—has been actively considered. The Law Commission for England and Wales recently published its recommendations for a scheme of remedies between separating cohabitants who had either had a child together or lived together beyond a minimum duration (fixed somewhere between two and five years), unless they had executed a valid 'opt-out agreement' excluding the operation of the scheme and (perhaps) made their own arrangements.⁵⁸ Under this scheme, the courts would have largely the same powers to reallocate assets as they enjoy on divorce, but those powers would be exercisable on a different basis. The Law Commission concluded that, given broadly observable differences between many cohabiting and marital relationships (including the substantial heterogeneity of the cohabiting population; the lack of express, public commitment to the relationship; and various patterns

⁵⁴ The courts also have exclusive jurisdiction (see Child Support Act 1991 s 8) to order periodical payments for meeting educational expenses, costs associated with extra needs created by a child's disability, and 'top-up' payments where the non-resident parent's income exceeds the jurisdictional limit of the statutory agency responsible for child maintenance: Children Act 1989 Sch 1; Matrimonial Causes Act 1973 s 23(1)(d) and (e). In the last case, in particular, payment may include a 'carer's allowance' to cover the parent with care's needs as carer: *Re P (A Child: Financial Provision)* [2003] EWCA Civ 837.

⁵⁵ These powers are exercisable by the courts under various statutes. In the case of married parents who are divorcing, jurisdiction would usually be exercised under the Matrimonial Causes Act 1973; in the case of unmarried parents, under the Children Act 1989 Sch 1.

⁵⁶ A v A [1995] 1 FCR 309.

⁵⁷ Most recently, Scotland: Family Law (Scotland) Act 2006.

⁵⁸ Law Commission (2007). See also Law Commission (2006) for its earlier consultation paper.

of money management and property ownership), provision between cohabitants should not be based on the needs of the parties or a norm of equal sharing. Instead, the court would exercise a principled discretion with restitutionary and compensatory purposes designed to respond to the enduring economic impact of the parties' contributions to their life together. The court would therefore aim broadly to reverse gains ('retained benefits') made by one party as a result of the other's contributions, such as the latter's direct or indirect financial contribution to the acquisition of property held in the former's sole name; and to share the burden of losses ('economic disadvantage') suffered by one party for the benefit of the family, for example, the loss of future earnings arising from having given up paid employment to care for children.

The Role of Empirical Data in Developing the Law

The consequence that, following Stack v Dowden, the proceeds of sale of the family home will be divided 50:50 in most joint ownership cases between cohabitants on separation may seem to resemble the approach in the context of ancillary relief in the wake of cases such as White v White and Miller: McFarlane. There is, however, a difference: the 'sharing principle' established in the latter context rests upon the view that the assets accrued during the 'partnership' of marriage should as a matter of principle be shared, while the presumption that equity follows the law in the context of the family home rests on the assumption that this is what such couples would usually have intended. The role that individual intentions play in the context of ancillary relief remains a moot point,⁵⁹ whereas evidence as to the parties' intentions may play a crucial role under the law of trusts.⁶⁰

This brings us on to the crucial distinction between empirically based presumptions and normative judgments. The principles developed and asserted in the recent House of Lords decisions may be put into these two inter-related categories:

- (i) empirically based presumptions about what parties to particular relationships may be taken to intend in particular circumstances: for example, that couples who purchase a home in joint names intend to hold it in equal shares; that couples who manage their money separately may not have that intention.
- (ii) normative judgements about what consequences ought to be taken to flow from particular situations: for example, that marriage is a partnership

⁵⁹ For example, pre-nuptial agreements retain a rather limited role, just one factor for the court to take into account in the exercise of its discretion, although in some cases an agreement may be a factor of 'magnetic importance': Crossley v Crossley [2007] EWCA Civ 1491.

⁶⁰ See eg Williamson v Sheikh [2008] EWCA Civ 990 (house in sole name of male partner, intentions gauged from an unexecuted declaration that female partner would be entitled to £10,000 plus 60% of proceeds).

which should entitle the partners to equal shares of certain property, regardless of need or financial contribution; that 'special contributions' should nevertheless receive recognition; that needs arising during a marriage (and perhaps to some extent afterwards) should generally be met by the other spouse; that some measure of 'compensation' ought to be afforded to recompense one spouse for economic disadvantage incurred, typically as a result of sacrificing paid employment to care for home and family.

To some extent, these empirically based presumptions and normative judgements articulated by the House of Lords were consciously based not just on the courts' assessment of the position of individual litigants who have appeared before them or on 'common sense' views adopted by the courts over time, but also on observations yielded from social science data, including socio-legal research. The speeches of Baroness Hale, in particular, are distinctive for and often enriched by her use of such material.⁶¹

Whether legal development is to be driven by reference to how people actually think or behave in their relationships, or by reference to what society considers the law ought to be seeking to achieve, or (perhaps most likely) a combination of both, social science data and analysis have central roles to play in informing the deliberations of appellate courts, law reformers and policy-makers. Such information is always essential to obtaining a proper understanding of the context in which the current law is operating and how that law is impacting on those to whom it applies. For example, data regarding participation in paid employment during relationships and about the economic disadvantage experienced following relationship breakdown by those who have been absent from the labour market may readily prompt the preliminary conclusion that remedies should be provided in response to this hardship.⁶² Determining precisely how that should be achieved, on what basis and in which cases may, of course, be more difficult to resolve, but research data may have a valuable contribution to make here too. In a rather different vein, data about how couples deal with the financial implications of separation, and about the obstacles that they encounter in the process, may prompt us to re-examine both procedural and other contextual aspects of dispute resolution and the extent to which complexity in the substantive law may itself impede settlement. Moreover, without a clear understanding of the social context in which the law is to operate, the law of unintended consequences is likely to rule the day.

⁶¹ In *Miller, McFarlane* and *Stack v Dowden* she refers to Pahl (1989), Ermisch (2000), Barlow et al (2001), Lewis (2001), Arthur et al (2002) and Douglas, Pearce and Woodward (2007) (*cf* the contribution by the latter authors to this volume (ch 7)). See also her extra-judicial writing: eg Hale (2004) draws on Rake (2000).

⁶² Although see remarks in the conclusion to this chapter regarding the relative responsibility of private family law and other areas of law and policy to deal with the issues explored in this volume.

The importance of empirical data can only grow as those considering legal reform in this field explore schemes based not on the exercise of a wholly individualised, strong discretion, with which many are becoming disenchanted, but on a more 'standardised' basis, such as community of property regimes, guidelines for wealth distribution, a community of property regimes, guidelines for wealth distribution, a community of the sort recently recommended for cohabitants by the Law Commission. If a (more or less) standardised scheme is to 'fit' the population to which it is to apply—to supply a satisfactory measure of 'average rather than individualised justice' supply a satisfactory measure of 'average rather than individualised justice' supply as: patterns of property ownership, money management, employment and so on, and what (if anything) these patterns imply about parties' intentions, commitment and so on; the degree of disparity in earning capacity or standard of living between partners following separation; and public attitudes towards the provision of remedies in particular cases and about the appropriate basis for remedies.

It is also important that the judges in our appellate courts should refer to social science data when developing the current law, as they have done in recent cases such as *Stack v Dowden*.⁶⁸ All too often their Lordships are not referred to key findings bearing on the empirical questions which they inevitably have to address in the course of their decision-making.⁶⁹ One major obstacle is, of course, the accessibility of such data: those presenting the case to the court will not usually be *au fait* with literature in non-legal disciplines, and even research assistants employed by the court may find it difficult to discover and interpret papers written by economists, sociologists and psychologists. It is equally unlikely, of course, that empirical researchers will be aware of ongoing legal proceedings to which their findings may be highly relevant, and terminological differences sometimes pose a barrier to mutual understanding. There is thus a role for intermediaries to make empirical research from one discipline more accessible—both physically and intellectually—to another.

It must, however, be conceded that no distillation of the existing research evidence will ever be value-free. Indeed, it is clear from academic criticisms of *Stack v*

⁶³ See eg the disadvantages of the current Matrimonial Causes Act scheme identified by the Law Society (2003) [71]; concerns reported in the wake of *Miller, McFarlane* by the Court of Appeal in *Charman v Charman (No 4)* [2007] EWCA Civ 503 [120]; and proposals for presumptive formulae for the calculation of periodical payments awards: eg Eekelaar (2006).

⁶⁴ See for example the Canadian spousal support guidelines: Department of Justice Canada (2008).

⁶⁵ Law Commission (2006, 2007).

⁶⁶ Rogerson and Thompson (2005) 10.

⁶⁷ Cf the view of Maclean and Eekelaar, in ch 2, this volume at p 34 regarding the difficulty today of identifying shared expectations and values on which reform might be based.

⁶⁸ See for example Harris-Short and Miles (2008), Probert (2008a, 2008b), and Douglas, Pearce and Woodward, ch 7, this volume.

⁶⁹ For example, there is no reference in *Stack v Dowden* to research into couples' money management practices, data which have been central to many commentators' criticisms of the case—see previous note.

Dowden that the research data cited in that case is open to interpretations different from those adopted by the House of Lords. Since such different interpretations can generate different conclusions about the appropriate response from the law, there is a need for constant evaluation of the interplay between such research evidence and judicial decision-making—an example of which appears in this volume, in chapter seven, by Douglas, Pearce and Woodward.

THE ISSUES AND THE DATA

As Maclean and Eekelaar indicate in chapter two of this volume, much research to date has focused on divorced or separating couples. But every separating couple was once an intact family, and the choices made while the relationship was subsisting will obviously be a significant constraint on the options available on relationship breakdown. Part II of the book, 'Work, Money and Property within Intimate Relationships: Expectations and Actions', examines data relevant to these questions.

Problems originating during the relationship are particularly acute where one or both of the parties are in debt. There has to date been relatively little discussion of this aspect of relationships or its legal implications: for example, we now recognise a loose concept of 'matrimonial property'—should we also recognise more explicitly the reality of 'matrimonial debt', and if so how should we distinguish it from personal debt?⁷⁰ Finney's discussion of debt within relationships and the links between debt and relationship breakdown in chapter six highlights key background data about both indebtedness and saving habits, and their association with particular types of relationship and life events such as relationship formation, the arrival of children, and relationship breakdown.

In addition, the House of Lords in both *Miller; McFarlane* and *Stack v Dowden* placed some importance on the way in which the couple dealt with their assets during the relationship, both when dividing assets on divorce and when discerning the parties' intentions regarding their property. But without empirical data it is difficult to know what forms of behaviour are unusual, or what light such behaviour can throw on the parties' intentions. The assumption that Mr Stack and Ms Dowden were unusual in keeping their finances separate is clearly contradicted by several studies. The chapters by Vogler (chapter four), Burgoyne and Sonnenberg (chapter five), and Douglas, Pearce and Woodward (chapter seven) cast valuable light on this area, showing in particular that what we might instinctively assume certain money management practices indicate about parties' intentions regarding ownership and use of their money might be incorrect.

Of course, an individual's behaviour may reflect constraints as well as their preferences, as is evident from the analysis by Scott and Dex in chapter three

⁷⁰ cf the findings of Cooke et al (2006), where respondents to their qualitative survey were largely not supportive of debts, as well as assets, being treated as joint as between spouses.

of the division of labour in the context of the UK labour market.⁷¹ Given the difficulties of changing the underlying conditions that produce such inequality, it is likely that it will persist for the foreseeable future, and it is therefore necessary to consider its effect both during relationships and upon separation. Unequal earnings may, for example, affect the way in which couples manage their money and generate differing concepts of 'equality' in this sphere, as chapter four by Vogler, and chapter five by Burgoyne and Sonnenberg show. For some, inequality in income is seen as mandating inequality in contributions, with each contributing according to their ability;⁷² for others, equal contributions are made despite differences in earnings. But as the next section shows, contributions, in whatever proportions, do not always translate into a share of the assets, equal or otherwise, upon relationship breakdown.

In the third part of the book, 'Dividing the Assets on Relationship Breakdown', the discussion of married and cohabiting couples of necessity diverges, owing to the very different legal regimes that apply to each on divorce and separation. Douglas, Pearce and Woodward in chapter seven draw on empirical research to highlight how difficult it is to ascertain the 'common intention' required by the law of trusts, particularly by reference to the factors identified by the House of Lords in *Stack v Dowden*. They show that the law has set itself a daunting task in seeking to give effect to parties' intentions, given the difficulties encountered in practice in attempting to ascertain from the parties' conduct of their relationship any shared intention specifically regarding property ownership.

Lewis, Tennant and Taylor in chapter eight identify those disadvantaged by the current law relating to cohabitants and consider whether the Law Commission's recommendations for reform in this area would do any better at addressing those individuals' problems. Their study shows, in particular, how legal ownership (specifically) rather than the needs of the parties (or beneficial entitlements under the law of trusts) currently dominates the way in which assets are divided when a cohabiting relationship breaks down. By contrast, upon divorce needs assume far greater importance: indeed, as Hitchings shows in chapter nine, the needs of the parties, and the range of practical options available to meet those needs, are of crucial importance in determining the nature of any settlement in most everyday cases. The emphasis on needs may help to achieve a degree of consistency in the disposal of these cases, despite the lack of guidance in the legislation. Further research is necessary in this area, not least given the concerns expressed by the Law Society in 2003 about uncertainty of outcome, which may vary between different county courts. 73 'Local knowledge' may be extremely important in practice, which poses a particular problem for the growing number of litigants in person appearing

⁷¹ And see further La Valle, Clery and Huerta (2008).

⁷² See also *Stack v Dowden* [2007] UKHL 17 [91], where Baroness Hale refers to the idea of 'each doing what they could', as a form of organisation that could then reflect an intention that the property be equally shared.

⁷³ Law Society (2003) [60]–[70]. See also Hitchings, ch 9, this volume.

before the family courts, who will almost certainly lack the insider knowledge garnered by repeat players.

Moreover, public misconceptions about the role of the law in this context may hinder a speedy settlement, and Dowding's chapter ten sets the issue of settlement in its practical context, identifying the many opportunities which parties have and are strongly encouraged by their legal advisors to use—to reach agreement without recourse to litigation, and the cost implications of their choices in this respect. As the chapters in this volume by Vogler (chapter four), Burgoyne and Sonnenberg (chapter five), and Douglas, Pearce and Woodward (chapter seven) indicate, couples often find it difficult to discuss money and property during their relationships—or easy to avoid having to do so. Inevitably if their relationship breaks down, such discussions become unavoidable and the earlier the couple can be helped to resolve their disagreements, the lower will be the costs of their dispute, both financial and emotional. But in the case of cohabitants (where there are no judicial powers of the sort enjoyed on divorce to adjust property rights and where the special matrimonial procedures described by Dowding in chapter ten do not apply), by the time the couple do belatedly address the issue of who owns what (and so who leaves the relationship with what), it may be too late for one party to guard against disadvantage of the sort discussed by Lewis, Tennant and Taylor in chapter eight.

One feature, however, is common to both cohabiting and married relationships, especially where there are dependent children, and that is inequality on relationship breakdown arising in particular from decisions made during the relationship about each party's participation in paid employment. The impact of these decisions on the income of men and women after divorce and the end of cohabitation is made manifest by Fisher and Low in chapter eleven, based on an analysis of several sweeps of the British Household Panel Survey. Their conclusion that income recovers over time gives some cause for optimism, although it is clear from their analysis that in many cases income recovery following divorce is achieved by repartnering rather than by improvements in personal income from employment. While this may guard against poverty following divorce, repartnering may be said simply to substitute a new dependency for the old one, and as such fails to address what for many is the underlying problem. Moreover, as Fisher and Low demonstrate, repartnering is less likely for older women and for women with children, leaving them economically disadvantaged unless able substantially to increase their labour market participation, which few appear to do.

Price's chapter on pensions (chapter twelve) builds on this theme of postdivorce inequality, providing powerful evidence that under current conditions divorced women do not and cannot catch up in pension building, and that, contrary to government's expectations, pension sharing on divorce is rare rather than routine, leaving divorced women facing poverty in old age. The reasons for the rarity of pension-sharing orders are hard to diagnose, although some anecdotal evidence suggests that pensions are not routinely included in parties' schedule of assets ('Form E') on divorce, so potentially rendering that valuable asset 'invisible' and not a 'matrimonial asset' to be shared. The assumption may be that women will create their own pension fund by 'downsizing' once the children reach independence, or that they will protect themselves via repartnering; and we know from research that some women are principally motivated to ensure their short-term security (for example in securing accommodation for themselves and the children) to the neglect of their long-term interests.⁷⁴ But as Fisher and Low's data show, the latter assumption may be misplaced, and the inherent uncertainty about prospects of repartnering and about the security of any new relationship makes it difficult to factor that possibility into any individual settlement. Moreover, the principles developed by the House of Lords in Miller; McFarlane have clear application to pension savings: savings accumulated during the marriage should be treated as matrimonial property and in principle be brought into account in dividing those assets pursuant to the 'sharing' principle; and women's disadvantage in relation to pensions offers a clear example of the sort of economic disadvantage to which the 'compensation' strand is intended to alert us. However, we also lack information about the characteristics of the tiny minority of cases in which such orders are made. This is clearly an area ripe for further investigation and, given the uncertainty that surrounds future state provision beyond retirement age, is a matter of considerable importance as the proportion of older women who are divorced, and living longer, increases.

Fisher and Low's data also provide material against which they evaluate alternative principles on which financial remedies following relationship breakdown might be based—comparing income-based measures (reflecting the impact of the relationship on each party's human capital) with consumption-based measures (based on needs or expected living standard)—and so for analysing the outcomes in and the incentive effects created by Miller; McFarlane. This theme—how to choose the most appropriate basis for financial relief—is developed further in the contrasting contributions of Dnes and Barlow in the final part of the book, 'A Rational Approach?'. Barlow in chapter fourteen draws on recent social attitude surveys and demographic changes to support her argument in favour of a 'bottom-up' approach to law reform: a reconfiguration of remedies on relationship breakdown according to whether the parties (married or not) had children, with a focus on needs-based remedies between couples with children and on greater autonomy for childless couples. Meanwhile, Dnes' chapter thirteen offers an economist's view of the law governing financial and property disputes on divorce and the end of cohabitation, demonstrating the potential for economic analysis to provide greater clarity to discussions in this area, for example by articulating welfare goals more explicitly and highlighting the different incentive effects of various settlement rules, including the introduction of financial remedies at the end of cohabitation.⁷⁵

⁷⁴ Douglas and Perry (2001).

⁷⁵ See generally Dnes and Rowthorn (2002).

These chapters also provide competing views on the ability of the law to influence behaviour within the family context, and how different research methodologies can be used to detect such influences at work. This discussion has been described by Barlow and others as being concerned with 'legal rationality', that is the idea that people respond to the framework of family regulation created by the law and order their intimate relationships in line with the law's expectations. ⁷⁶ Barlow argues, in light of data yielded from various research projects conducted by herself and other socio-legal scholars, that legislators in the family-law sphere are in danger of making a 'rationality mistake', overestimating law's ability to influence behaviour in the family field. She considers the evidence of people's changing attitudes to relationships and property and how well or otherwise this fits with the law's expectations, arguing that the legal and financial consequences of marriage rarely figure in the decision whether to marry (at least for first marriages). Even assuming that the parties have an accurate understanding of the law, to marry for such pragmatic reasons is seen by many as distasteful, despite the potentially far-reaching economic consequences of this decision when the relationship ends, whether by separation or by death. Dnes, by contrast, explains the insights that can be provided by positivistic, quantitative analysis of longitudinal samples (such as that undertaken by Fisher and Low), arguing that how people explain their own behaviour to qualitative researchers may not necessarily correspond with independent indicators about what people are in fact doing, and highlighting a number of studies identifying significant changes in certain types of behaviour, including partnering behaviour, in response to legal change. This debate about the influence of law on family behaviour is undoubtedly one that will continue and requires further examination. It may be that some laws, such as those relating to welfare benefits and tax credits⁷⁷ have greater influence over family practices than those relating to financial settlements following relationship breakdown, 78 not least given what to the individuals concerned is the relative complexity and perceived remoteness of those laws from decisions about marriage and conduct within ongoing relationships.

WHERE SHOULD WE GO FROM HERE?

One clear finding that emerges from the empirical research is the great diversity of practices within families. Whether such diversity militates against devising proposals for reform on the basis of shared values, as Maclean and Eekelaar

⁷⁶ In the language of economists, 'rationality' is simply concerned with observations about the consistency or otherwise of a subject's decisions, and is entirely unrelated to the desirability of those decisions, the information on which they are based, or their compatibility with or conformity with a particular policy goal or objective perception of the subject's best interests, given the current legal framework.

⁷⁷ See eg Anderberg (2008).

⁷⁸ Although see Mechoulan (2006), who found that the rules governing allocation of property on divorce had a stronger effect on the divorce rate than the grounds for divorce.

suggest in chapter two, is a moot point. The surveys discussed by Barlow in chapter fourteen suggest that there is still a considerable degree of common ground in the wider population, and a solution that does not resonate with individuals' values may be difficult to implement in practice.

As noted above, there are difficulties in ascertaining the parties' intentions regarding their property from their money management practices during their relationship. There is also a more fundamental question to be posed, which is how far the law *should* be guided by the parties' intentions when dividing assets at the end of a relationship. To date, as we have seen, the answer to that question has differed according to the nature of the relationship, rather than the extent of the disadvantage experienced by the individual in question. This has resulted in a situation in which very similar contributions may result in very different outcomes (as demonstrated by Lewis, Tennant and Taylor, in chapter eight, who compare the results that their cohabitants might have expected had they been married with the actual arrangements that were made).

Yet although individual experiences differ, some common trends can be observed. While we have, as a society, departed from the pure version of Lloyd Cohen's profile of marriage, 79 it remains the case that women tend to earn less than men and spend more time on unpaid household tasks. Whether the first causes the second, or vice versa, is difficult to determine; still more difficult is the task of devising a solution to encourage greater equality in both spheres.⁸⁰ The positive story of women's enhanced earning capacity and labour market participation has a negative consequence: the extent of downward mobility for those who move out of the labour market even for a short period is greater for today's women than it was for previous generations. And although remaining out of the labour market to care for children and other dependants should be respected as a choice, this does not mean that the financial detriment of doing so should fall on the carer alone. A division of responsibilities that works within the intact family may well leave the partner who has not pursued a career vulnerable on relationship breakdown (as demonstrated by Fisher and Low and by Price in chapters eleven and twelve).

To the extent that the other party—whether married or cohabiting—has benefited from the contribution of the one who provided care or other non-tangible contributions, then there is a justification for the court to intervene as between the parties themselves. But on what basis? Underlying much of the discussion in this volume is the concept of 'equality'. But this is a multi-faceted concept which could support a variety of solutions to the problems experienced on relationship breakdown. Should equal sharing of assets arise only following equality of contribution? Is contribution to be regarded as equal only where it is of equal value,

⁷⁹ Cohen (1987).

⁸⁰ See eg recent comments of Nicola Brewer, chief executive of the Equalities and Human Rights Commission, regarding the negative impact that improved maternity rights have had on women's progression in the labour market: Bennett and Ahmed (2008).